Author’s response to reviews

Title: «Intention to use» electronic medical record and its predictors among health care providers at referral Hospitals, north-west Ethiopia, 2019: using Unified Theory of Acceptance and Use Technology 2(UTAUT2) model.

Authors:

Mohammedjud Hassen Ahmed (judhassen423@gmail.com)
Adina Demissie Bogale (adinabogale@gmail.co)
Binyam Chaklu Tilahun (binigcms@gmail.co)
Mulugeta Haylom Kalayou (mhayelom5@yahoo.com)
jorn Klein (Jorn.Klein@yahoo.no)
Shegaw Anagaw Mengiste (Shegaw.Mengiste@un.no)
Berhanu Fikadie Endehabtu (berhanufikadie@gmail.co)

Version: 4 Date: 12 Aug 2020

Author’s response to reviews:

Dear Editor, i have tried to incorporate your comments based on your request. Thank you so much for teaching me some sciences regarding to research